Author’s response to reviews

Title: Effect of 6-months of physical exercise on the nitrate/nitrite levels in hypertensive postmenopausal women

Authors:

Pedro R Zaros (przaros@uol.com.br)
Carla EMR Pires (carlapires@hotmail.com)
Mauricio Bacci Jr (mbacci@rc.unesp.br)
Camila Moraes (mormila@hotmail.com)
Angelina Zanesco (azanesco@rc.unesp.br)

Version: 2 Date: 28 April 2009

Author’s response to reviews: see over
April, 28th 2009

Dear Professor
Melissa Norton, MD
Editor-in-chief of BMC Women’s Health

Dear Editor-in-chief

Please find attached the revised manuscript entitled: Effect of 6-months of physical exercise on the nitrate/nitrite levels in hypertensive postmenopausal women by Zaros et al. to be resubmitted to the BMC Women’s Health.

I am looking forward hearing you soon.

My best regards

Angelina Zanesco, PhD
Department of Physical Education
Av 24^A, 1515
13506-900
Rio Claro (SP), Brazil.
Fax: +55 19 35264321
Phone: +55 19 35264324
e-mail: azanesco@rc.unesp.br